Author's response to reviews

Title: Palliative care for the elderly - developing a curriculum for nursing and medical students.

Authors:

Johannes M Just (johannes.just@uni-wh.de)
Christian Schulz (christian.schulz@uni-wh.de)
Maren Bongartz (maren.bongartz@uni-wh.de)
Martin W Schnell (schnell@uni-wh.de)

Version: 5 Date: 14 September 2010

Author's response to reviews:

Dear editors,

you have asked for the following revisions:

Major revisions (we require the author to make these changes)
References: reference 35, Submitted/ unpublished. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, submitted

I HAVE CHANGED THE CITATION AS INDICATED

Tables: Eight tables have been uploaded by only five have been cited. Either delete the extra tables or cite them in the text.

THE TABLES HAVE BEEN CITED (PAGE 7, "RESULTS STAGE FOUR")

References: please cite all references in numerical order in the text. References 21-25 have not been cited in the correct position.

THE REFERENCES ARE NOW EMBEDED IN THE MAIN TEXT.

Best regards,

Johannes Just